Author's response to reviews

Title: Evaluation of the Widal tube agglutination test for the diagnosis of typhoid fever among children admitted to a rural hospital in Tanzania and a comparison with previous studies

Authors:

Benedikt Ley (bnley@ivi.int)
George Mtove (mtoveq2002@yahoo.co.uk)
Kamala Thriemer (kthriemer@ivi.int)
Ben Amos (bennsal@teule.org.uk)
Lorenz von Seidlein (lseidlein@gmail.com)
Ilse Hendriksen (icehendriksen@hotmail.com)
Abraham Mwambuli (mwambuli1978@yahoo.com)
Aikande Shoo (muhezaddh@kaributanga.com)
Rajabu Malahiyo (mallahiyo@hotmail.com)
Shaali M. Ame (shaaliame@yahoo.com)
Deok R. Kim (dkim@ivi.int)
Leon R. Ochiai (rlochiai@ivi.int)
John D. Clemens (jclemens@ivi.int)
Hugh Reyburn (Hugh.Reyburn@lshtm.ac.uk)
Harald Wilfing (harald.wilfing@univie.ac.at)
Stephen Magesa (smagesa@hotmail.com)
Jacqueline L. Deen (jdeen@gmail.com)

Version: 6 Date: 17 June 2010

Author's response to reviews:

Corrected authors names, they are now written identical to the way they have been entered in the system.